Author's response to reviews

Title: Injuries at a Canadian National Taekwondo Championships

Authors:

Mohsen Kazemi (mkazemi@cmcc.ca)
Willy Pieter (yshin516@yahoo.com)

Version: 3 Date: 6 July 2004

Author's response to reviews:

July, 4, 2004

Dear Referees and editors,

We have included the confidence intervals requested by Dr. Cassidy and hope that this is satisfactory. We have also included the reference for our definition of "Injury" for Dr. Burke. With regards to "Joint Dysfunction", we believe that we have included sufficient references and explanation, which have satisfied Dr. Cassidy who is a well published author in this area. In addition, the following recently published research article, "An investigation into the validity of cervical spine motion palpation using subjects with congenital block vertebrae as a 'gold standard'" by Humphreys BKIM, Delahaye M, Peterson CK, BMC Musculoskeletal Disorders 2004, 5:19, may shed more light on joint dysfunction.

We hope that our response is satisfactory.

Sincerely,

Mohsen Kazemi and Willy Pieter